Author's response to reviews

Title: Variability in the performance of preventive services and in the degree of control of identified health problems: A primary care study protocol.

Authors:

Bonaventura Bolíbar (bbolibar@idiapjgol.org)
Clara Pareja (30193cpr@comb.es)
María Pilar Astier-Peña (mpastier@salud.aragon.es)
Julio Morán (jmoranpi@cfnavarra.es)
Teresa Rodríguez-Blanco (trodriguez@idiapjgol.org)
Magdalena Rosell-Murphy (mrosell@idiapjgol.org)
Manuel Iglesias (mrodal@menta.net)
Sebastián Juncosa (sjuncosa@sapcll.scs.es)
Juanjo Mascort (jmascort@semfyc.es)
Concepció Violan (cviolan@idiapjgol.org)
Rosa Magallón (med000764@gmail.com)
Javier Apezteguia (japezteu@cfnavarra.es)

Version: 3 Date: 7 August 2008

Author's response to reviews:

Dear Editor,

Please find attached our revised manuscript entitled “Variability in the performance of preventive services and in the degree of control of identified health problems: A primary care study protocol”.

In this revision we have list authors below the title, we have included the corresponding author wit his e-mail, we have removed the fixed footnote and we have made a final check of the manuscript.

Yours sincerely,

Bonaventura Bolíbar